Author's response to reviews

Title: Length of stay and associated costs of obesity related hospital admissions in Ireland

Authors:

Akke Vellinga (akke.vellinga@nuigalway.ie)
Diarmuid O'Donovan (diarmuid.odonovan@nuigalway.ie)
Davida DelaHarpe (davida.delaharpe@mailf.hse.ie)

Version: 4 Date: 11 April 2008

Author's response to reviews:

Editorial changes were made to the references and the two figures were uploaded seperately.